Author's response to reviews

Title: The positive association between number of children and obesity in Iranian women and men: Results from the National Health Survey

Authors:

Enayatollah Bakhshi (Bakhshi@razi.tums.ac.ir)
Mohammad Reza Eshraghian (eshraghian@yahoo.com)
Kazem Mohammad (mohamadk@tums.ac.ir)
Abbas Rahimi Foroughi (rahimifo@tums.ac.ir)
Hojat Zeraati (zeraatih@tums.ac.ir)
Akbar Fotouhi (afotouhi@tums.ac.ir)
Fraidon Siassi (siassif@sph.tums.ac.ir)
Behjat Seifi (behins@yahoo.com)

Version: 10 Date: 24 May 2008

Author's response to reviews:

Dear Natalie Pafitis,
Assistant Editor
Thanks for accepting our manuscript for publication in your valuable journal. Please find here the final version of our manuscript ready for publication.
Sincerely Yours
M. R. Eshraghian
Corresponding Author